\begin{table}%t2
\caption{Data used in this work (continued).}
%\centering
{\small
\begin{tabular}{lcccccc}
\hline\hline
Source &    $l$ &   $b$    & \EBV\tablefootmark{a} & \tauhi\tablefootmark{b} & \tauhcop\tablefootmark{c} & \WCO\tablefootmark{d} \\
       & $\circ$ & $\circ$ & mag      & \kms       & \kms &      \Kkms \\
\hline
   B0056+666& 123.782& 3.992& 1.20& 13.30(1.05)& &  \\
   J0102+584& 124.426& $-4.436$& 0.56& 9.75(0.10)& 0.34(0.01)& $<$0.34 \\
   B0107+562& 125.637& $-6.231$& 0.39& 4.56(0.07)& &  \\
   B0125+628& 127.109& 0.538& 1.60& 36.54(0.59)& & 6.81(0.09) \\
   B0212+735& 128.927& 11.964& 0.76& 12.36(0.13)& 4.98(0.20)& 6.81(0.06) \\
   B0205+614& 132.064& 0.210& 1.55& 27.59(0.45)& & 21.54(0.06) \\
   B0224+671& 132.122& 6.234& 1.00& 14.73(0.24)& 2.46(0.07)& 2.27(0.06) \\
   B0241+623& 135.636& 2.431& 0.73& 8.38(0.71)& &  \\
   B0323+55A& 143.890& $-1.057$& 1.75& 37.96(0.30)& &  \\
   B0300+471& 144.986& $-9.863$& 0.25& 4.99(0.14)& &  \\
   B0954+658& 145.746& 43.132& 0.12& & 1.23(0.29)& 1.94(0.04) \\
   B0332+534& 145.952& $-1.681$0& 1.68& 31.89(0.35)& &  \\
   B0334+506& 147.809& $-3.895$& 1.24& 18.77(0.14)& &  \\
   B0430+587& 148.581& 7.536& 0.54& 5.80(0.19)& &  \\
   B0355+508& 150.377& $-1.604$& 1.50& 42.79(1.10)& 6.48(0.03)& 16.82(0.12) \\
        3C84& 150.577& $-13.261$& 0.05& & $<$0.10& $<$0.20 \\
   B0442+506& 155.877& 3.460& 0.94& 13.14(0.21)& &  \\
   B0435+487& 156.417& 1.317& 1.35& 27.84(0.22)& &  \\
   B0235+164& 156.772& $-39.108$& 0.03& & $<$0.20& $<$0.20 \\
   B0404+429& 156.780& $-6.586$& 0.49& 11.73(0.13)& &  \\
   B0420+417& 159.705& $-5.382$& 0.74& 12.93(0.12)& &  \\
   B0458+477& 159.712& 3.653& 0.72& 15.07(0.14)& &  \\
   B0406+386& 159.845& $-9.484$& 1.02& 12.12(0.12)& &  \\
   B0429+416& 160.965& $-4.342$& 0.56& 11.55(0.04)& &  \\
   B0415+379& 161.686& $-8.788$& 1.66& 12.14(0.14)& 13.34(0.65)& 17.45(0.11) \\
   B0442+39A& 164.109& $-3.656$& 0.56& 6.69(0.12)& &  \\
   B0509+406& 166.502& 0.916& 0.53& 14.45(0.16)& &  \\
   B0552+398& 171.647& 7.285& 0.44& 7.71(0.12)& 0.69(0.08)& 0.33(0.04) \\
   B0923+392& 183.709& 46.165& 0.10& & $<$0.09& $<$0.20 \\
   B0601+204& 189.566& $-0.640$& 1.37& 33.58(0.56)& &  \\
   B0528+134& 191.368& $-11.012$& 0.89& 11.88(0.20)& 2.14(0.02)& 2.53(0.06) \\
   B0629+160& 196.582& 3.204& 0.46& 7.54(0.25)& &  \\
   B0622+147& 196.983& 1.103& 0.82& 26.07(0.46)& &  \\
   B0629+104& 201.531& 0.508& 1.81& 27.96(0.25)& &  \\
   B0630+082& 203.544& $-0.272$& 0.91& 28.65(0.33)& &  \\
   B0624-058& 215.439& $-8.067$& 0.71& 5.55(0.02)& &  \\
   J0008+686& 215.752& $-13.253$& 1.28& & 7.16(0.75)& 13.79(0.05) \\
   B0605-085& 215.752& $-13.523$& 0.58& & 2.17(0.25)& 1.31(0.13) \\
   B0736+017& 216.990& 11.380& 0.13& 2.96(0.17)& 0.89(0.10)& 0.92(0.04) \\
   B0607-157& 222.611& $-16.183$& 0.26& & 0.36(0.09)& $<$0.20 \\
   B0727-115& 227.768& 3.140& 0.30& 6.81(0.14)& 0.55(0.02)& 0.81(0.04) \\
   B0733-174& 233.585& 1.444& 0.92& 20.00(0.13)& &  \\
   B0709-206& 233.670& $-5.021$& 0.98& 19.22(0.28)& &  \\
   B0704-231& 235.337& $-7.218$& 0.33& 7.28(0.10)& &  \\
   B1055+018& 251.513& 52.775& 0.20& & $<$0.24& $<$0.20 \\
       3C273& 289.954& 64.360& 0.03& & $<$0.08& $<$0.20 \\
       3C279& 305.107& 57.062& 0.05& & $<$0.07& $<$0.20 \\
   B1334-127& 320.026& 48.374& 0.04& & $<$0.17& $<$0.20 \\
   B1714-397& 347.748& $-1.142$& 3.53& 39.45(0.20)& &  \\
   B1705-353& 350.339& 2.768& 1.74& 21.39(1.30)& &  \\
   B1714-336& 352.735& 2.393& 2.42& 29.58(0.88)& &  \\
   B1711-285& 356.516& 5.884& 0.60& 8.64(0.27)& &  \\
\hline
\end{tabular}}
\tablefoot{\tablefoottext{a}{Reddening from \cite{SchFin+98} rms error is 16\%;}
\tablefoottext{b}{\tauhi\ from \cite{GarDic89} and (for J-sources) this work see Sect.~2;}
\tablefoottext{c}{\tauhcop\ from \cite{LucLis96}, \cite{LisLuc00} and this work see Sect.~2;}
\tablefoottext{d}{\WCO\ from \cite{LucLis96}, \cite{LisLuc98} and this work see Sect.~2.}}
\end{table}